Author's response to reviews

Title: Solitary breast metastasis from myxoid liposarcoma.

Authors:

Masahiro Yokouchi (masahiro@m3.kufm.kagoshima-u.ac.jp)
Satoshi Nagano (naga@m2.kufm.kagoshima-u.ac.jp)
Yuko Kijima (ykijima@m3.kufm.kagoshima-u.ac.jp)
Heiji Yoshinaka (heiji@m3.kufm.kagoshima-u.ac.jp)
Hirofumi Shimada (mac1mickmick2@yahoo.co.jp)
Shunsuke Nakamura (hhxt0y@hotmail.co.jp)
Takako Yoshioka (yoshioka@m2.kufm.kagoshima-u.ac.jp)
Akihide Tanimoto (akit09@m3.kufm.kagoshima-u.ac.jp)
Shoji Natsugoe (natsugoe@m2.kufm.kagoshima-u.ac.jp)
Setsuro Komiya (skomiya@m3.kufm.kagoshima-u.ac.jp)

Version: 2  Date: 28 April 2014

Author's response to reviews:

Dear Editor:

On behalf of all the authors, I request you to consider our manuscript entitled “Solitary breast metastasis from myxoid liposarcoma.” for publication in BMC Surgery as a case report.

Metastasis of soft tissue sarcoma to the breast is very rare. In this report, we present an extremely rare case of a solitary metastatic breast tumor arising from myxoid liposarcoma of the lower limbs. Five years following resection of the metastatic breast lesion, the patient is without evidence of disease and has a good quality of life.

To the best of our knowledge, no isolated case reports of solitary breast metastasis by myxoid liposarcoma have been previously reported in the English-language literature.

We believe that the findings reported will be of special interest to the readers of BMC Surgery.

This manuscript has not been published and is not under consideration for publication elsewhere. All the authors have read the manuscript and approved this submission.

Competing interests
The authors declare that they have no competing interests.

Ethical Adherence

This study was performed in accordance with the Helsinki Declaration and the written consent by the patient.

Sincerely,

Masahiro Yokouchi, MD, PhD
Department of Orthopedic Surgery
Kagoshima Graduate School of Medical and Dental Sciences
8-35-1 Sakuragaoka
Kagoshima, Japan 890-8520
Tel. +81-99-275-5381
Fax +81-99-265-4699
E-mail: masahiro@m3.kufm.kagoshima-u.ac.jp